Author's response to reviews

Title: High School Suicide in South Africa: Teachers' knowledge, views and training needs

Authors:

Hilda N Shilubane (hilda.shilubane@univen.ac.za)
Arjan E.R Bos (arjan.bos@ou.nl)
Robert A.C Ruiter (r.ruiter@maastrichtuniversity.nl)
Bart van den Borne (b.vborne@maastrichtuniversity.nl)
Priscilla S Reddy (preddy@hsr.ac.za)

Version: 2 Date: 24 October 2014

Author's response to reviews: see over
Dear Editor,

Enclosed you will find a revised version of the manuscript “High School Suicide in South Africa: Teachers’ knowledge, views and training needs” for possible publication in BMC Public Health. The manuscript is co-authored by Arjan E.R Bos, Robert A.C. Ruiter, Bart H.W van den Borne and Priscilla S. Reddy.

The abstract contains background, methods, results and conclusion. The article sections are aligned with the journal requirements. The manuscript is numbered for easy reference.

Hope to hear from you soon.

Yours sincerely,

Hilda Shilubane, PhD